Author’s response to reviews

Title: A protein interaction map identifies existing drugs targeting SARS-CoV-2

Authors:

claudia cava (claudia.cava@ibfm.cnr.it)
Gloria Bertoli (gloria.bertoli@ibfm.cnr.it)
Isabella Castiglioni (isabella.castiglioni@ibfm.cnr.it)

Version: 1 Date: 01 Jun 2020

Author’s response to reviews:

Dear editor,

we moved the Declaration section before the Reference.

Kind regards